Author’s response to reviews

Title: The effect of Liv-52 on liver ischemia reperfusion damage in rats

Authors:
Orhan Cimen (orhancimen24@hotmail.com)
Huseyin Eken (huseyineken80@hotmail.com)
Ferda Keskin Cİmen (ferdakeskinn@hotmail.com)
Arif Burak Cekic (abcekcic@hotmail.com)
Nezahat Kurt (nezahatkurt@hotmail.com)
Asli Ozbek Bilgin (asliozbekbilgin@gmail.com)
Bahadir Suleyman (bahadirsuleyman@yandex.com)
Halis Suleyman (halis.suleyman@gmail.com)
Renad Mammadov (renad_mamedov@hotmail.com)
Kamil Pehlivanoglu (pehlivan550@gmail.com)
Eray Kurnaz (erakurnaz@hotmail.com)

Version: 4 Date: 13 Dec 2019

Author’s response to reviews:

Dear editor,

We made the changes that you recommended.

Sincerely…

Editor Comments:

1. Thank you for explaining the way tablets were prepared for oral gavage. For clarity, your response “In our study, Liv-52 was given to the LIR animal group orally by means of a catheter at a dose of 20 mg/kg and volume of 1 cc. 20 mg for 1000 mg rat, each rat weighs between 250-270 mg. One tablet weighing 320 mg dissolved in 64 cc distilled water and 1 cc given by gavage orally. 1cc has 5 mg Liv52(20mg/kg) and also contains as one tablet (weighing 320 mg) of Liv52 contains; Capparis Spinosa %20,3125, Cichorium Intybus %20,3125, Mandur Bhasma %10,3125, Solanum Nigrum %10, Terminalia Arjuna %10,Cassia Occidentalis %5, Achillea Millefolium %5 and
Tamarix Gallica % 5.” should be included in the main text of the manuscript (methods), in case someone wants to replicate the experiments.
Response: Preparation of the tablets for oral gavage explained in the manuscript, Methodology section, experimental procedure, line 7.